Author's response to reviews

Title: A multifaceted intervention to implement guideline care and improve quality of care for older people who present to the emergency department with falls

Authors:

Nicholas Waldron (ngwaldron@gmail.com)
Ian Dey (Ian.Dey@health.wa.gov.au)
Yusuf Nagree (Yusuf.Nagree@health.wa.gov.au)
Jianguo Xiao (p_xiao@bigpond.com)
Leon Flicker (Leon.Flicker@uwa.edu.au)

Version: 3 Date: 24 January 2011

Author's response to reviews: see over
Cover Letter

To The BMC Series Editorial Production Team

Re – “A multi-faceted intervention to implement guideline care and improve quality of care for older people who present to the emergency department with falls” MS ID : 1209482367441095

Thanks you for you very helpful review and suggested changes. I have addressed the major revision and partially addressed the minor revision as my skill base allows. Below I have outlined what I have and have not done.

Major revisions

1) Authors' contributions: Please use initials instead of full name.
   - Corrected as advised

2) Authors' contributions: Please indicate that all authors read and approved the final manuscript.
   - Corrected as advised

3) Tables: You have included a table on pages 18 - 20 of your manuscript. Please either cite this as a table throughout the manuscript, remove the lines and include it as text or remove it from the manuscript.
   - Table removed

4) References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example 'text [1].'
   - Corrected as advised

Minor revisions

1) Affiliations: Please remember to include the city and the country in the affiliation details.
   - I have corrected this as I understand removing the state Western Australia and replacing the country. I hope this is what is meant.

2) Funding: Please move the statement of funding to the Acknowledgements section and remove it from the title page.
   - Corrected as advised

3) Abstract: Please insert the heading 'Abstract' above the abstract.
   - Corrected as advised

4) Introduction: Please rename this section 'Background'.
   - Corrected as advised

5) Acknowledgements: Please insert this section directly after the authors' contributions.
   - Corrected as advised
6) Conflict of interest: Please rename this section 'Competing Interests'.
   - Corrected as advised

7) Sponsor's Role: Please insert this section in the Acknowledgements.
   - Corrected as advised

8) References: please provide the full citation details for reference 4. You can view the BMC reference guide at the following link http://www.biomedcentral.com/bmcgenet/ifora/#references.
   - I have not done this.

8) Please remove any blank pages from the manuscript.
   - Corrected as advised

9) Figures: The image file should not include the title (e.g. Figure 1... etc.) or figure number. The legend and title should be part of the manuscript file after the reference list. The figures are numbered automatically in the order in which they are uploaded.
   - I have not done this.

10) Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures.
   - I have not done this.

11) Tables- please remove the visible vertical lines from your tables.
   - I have not done this.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
   - I have checked for errors and made minor changes.

Thankyou for your kind attention to this manuscript it is much appreciated.

Dr Nicholas Waldron